Pbar Department Meeting



3-18-04

Machine Coordinator:   Elvin Harms

Status of Shutdown:


Elvin goes over the completed work list.

Ongoing


Flying wires: Vladimir



All hooked up but one - Core vertical – problems.



All other wires fly normally



Acnet interfaces ready


BLMs:



Needs adjustment


LLRF board – Bill A.


Mark D. -  quad stands


Budlong – 


Obie – Debuncher quad shunts hooked up via Keith’s memo



D:IB, A:IB done- A:IB 


DVM – 



Rad survey of Debuncher change



End of AP2 hot



DRF2 cooled off, other things around it hotter


19 quads in arcs hot.


Keith


Accumulator BPM fixed 



H102 has a name plate of H103


Tony



Test lens 29 – conductivity – 




Preload may be wrong

Want to understand what is wrong with lens 29 before they assemble and fill lens 30



Want to run at lower gradient when we turn back on

Schedule


See web schedule

Startup Studies


Pre-beam



Site risers



Injectors start up



Secure pbar



PS turnon




IB regulation


120 Gev startup



P1 rolls



Deb cooling



Stacking tuneup



Deb mom cooling characterization


8 gev startup



P1 rolls



Check aperatures



Deb quad functions


Stacktail notch filters – 8hrs, no stacking


Studies



Accumulator energy



Debuncher energy

LBNL

Static eta 
Tank centering

DRF2

Debuncher quad centering

Deb bumps

AP2 BPMs

Paint the aperature

